Author’s response to reviews

Title: Trends in Comprehensive Abortion Care (CAC) and Characteristics of Women Receiving abortion care in Tertiary Hospital in Nepal.

Authors:

Jamuna Sayami (jamunats10@gmail.com)

Version: 1 Date: 03 Sep 2017

Author’s response to reviews:

List of abbreviations:

CAC: Comprehensive Abortion Care
THTH: Tribhuvan University Teaching Hospital
MVA: Manual Vaccum Aspiration
MA: Medical abortion
PAC: Post Abortion Care
POC: Product of Conception
WHO: World Health Orgnization

Ethical approval:

Ethical approval was obtained from Institutional Review board of Institute of medicine, TU, Nepal.

Competing Interest:

There is no competing interest. The author is solely responsible for the manuscript.
Funding source:

There is no funding available for the study. It is done by authored voluntary contributions.

Authors Contributions:

All data collection, analysis, manuscript writing obtaining ethical approval.